Author's response to reviews

Title: Human papillomavirus (HPV) types 16, 18, 31, 45 DNA loads and HPV-16 integration in persistent and transient infections in young women

Authors:

Agnihotram V. Ramanakumar (venkata.agnihotram@mcgill.ca)
Otelinda Goncalves (goncalves.otelinda.chum@ssss.gouv.qc.ca)
Harriet Richardson (HRichardson@ctg.queensu.ca)
Pierre Tellier (pierre-paul.tellier@mcgill.ca)
Alex Ferenczy (alex.ferenczy@mcgill.ca)
François Coutlée (francois.coutlee.chum@ssss.gouv.qc.ca)
Eduardo L Franco (eduardo.franco@mcgill.ca)

Version: 6 Date: 8 November 2010

Author's response to reviews:

Dear editorial production team,

We acted on your comments and revised the references according to your previous mail. We also corrected the way we should quote them in the manuscript.

Thanks,
authors